Author’s response to reviews

Title: A girl with protein-losing enteropathy during ketogenic diet: a case report

Authors:

juan wang (wangjuannn@126.com)
Li Jiang (lijiangcmu@126.com)
Min Cheng (836825567@qq.com)

Version: 1 Date: 30 Jun 2019

Author’s response to reviews:

1. Written informed consent has obtained from the patient’s parent for the publication of this Case Report.

2. The CARE checklist has revised.

3. The title page has revised.

4. The declarations has revised.

5. Figures and additional files have uploaded by separate files.